Author's response to reviews

Title: Per oral Substitution with 300000 IE Vitamin D Reduces Bone Turnover Markers and Increases 25-OH Vitamin D Levels in HIV-positive Patients

Authors:

Rein Jan Piso (jan.piso@spital.so.ch)
Madeleine Rothen (Madeleine.Rothen@spital-biel.ch)
Jean Pierre Rothen (jp.rothen@aarelab.ch)
Matthias Stahl (matthias.stahl@spital.so.ch)
Christoph Fux (christoph.fux@sec.ksa.ch)

Version: 2 Date: 23 September 2013

Author's response to reviews:

We have added the details concerning ethic approval, competing interests, authors contributions and acknowledgement, which were missing in the primary version.